Author's response to reviews

Title: Excess of transmission of the G allele of the -1438A/G polymorphism of the 5-HT2A receptor gene in patients with schizophrenia responsive to antipsychotics.

Authors:

Dalila Benmessaoud (ben.dalila@gmail.com)
Nora Hamdani (nora.hamdani@lmr.aphp.fr)
Claudette Boni (claudette.boni@bichat.inserm.fr)
Nicolas Ramoz (nicolas.ramoz@bichat.inserm.fr)
Michel Hamon (hamon@ext.jussieu.fr)
Farid Kacha (f.kacha@caramail.com)
Philip Gorwood (philip.gorwood@lmr.aphp.fr)

Version: 2 Date: 30 January 2008

Author's response to reviews:

A letter detailing the proposed responses to reviewers is attached.